Author's response to reviews

**Title:** Elective high-frequency oscillatory ventilation in preterm infants with respiratory distress syndrome: an individual patient data meta-analysis.

**Authors:**

Filip Cools (filip.cools@uzbrussel.be)
Lisa M Askie (laskie@ctc.usyd.edu.au)
Martin Offringa (m.offringa@amc.uva.nl)
PreVILIG Collaboration Prevention of Ventilator Induced Lung Injury Study Group (filip.cools@uzbrussel.be)

**Version:** 3 **Date:** 13 May 2009

**Author's response to reviews:** see over
Major revisions (We made these changes)

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. **DONE**

Author groups - If you wish to include the name of the group in the list of authors on the title page, you must remove the names of individual members of the group from the authors list in the submission system and the manuscript. You must also add the name of the group to the first name box in the submission system, and add the abbreviation of the name in brackets to the last name box. All the members of the group should then be listed in the Acknowledgments section. **DONE**

Alternatively, you may include the names of all authors from the group in the list of authors on the title page. We chose to include the name of the group in the list of authors. If you do this, you must also include all the authors in the submission system. You should omit the name of the group from the list of authors on the title page and the submission system, but you should add a note in the Acknowledgements listing the members of the group, and describe the role of the group in this study in the authors contributions section. **DONE**

References - we cannot currently find reference 6 and 18 cited in the manuscript text. Please ensure that all the references are cited in the text in numerical order. Any references not cited in the text will not be included in the final publication. **All the references are cited in the text in numerical order.**

Minor revisions (We made these changes as well).

REVISED VERSION, Study protocol - please remove this text from the title page. **DONE**

Email addresses - Please use the following format- **DONE on title page – is impossible to do in the submission system.**
GSF: g.s.feder@qmul.ac.uk
HS: hsohal@yahoo.co.uk
SE: s.eldridge@qmul.ac.uk

Reference citations - please remove the squares from in front of every reference citation within the text. **DONE**

Highlighting - please remove this from the manuscript. **DONE**

Underlining - Please note that underlining is not supported in the final full text; you may wish to change it to bold. **DONE**

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance. **DONE**

Thank you!